

Response to Review

- Add a short running title of less than 6 words.
 - See annotated document. Thank you.
- Please add authors.
 - See annotated document. Thank you.
- The ethic approval documents/letters must be provided in PDF format and each statement must also be mentioned as a footnote in the manuscript text.
 - See annotated document and see attached PDFs. Thank you.
- Please provide your address, telephone, and fax
 - See annotated document. Thank you.
- Abstract
 - See annotated document. Thank you.
- Please list 5–10 key words for each paper, selected mainly from Index Medicus, which reflect the content of the study. Each key word is separated by a semicolon.
 - See annotated document. Thank you.
- Please write a summary of less than 100 words to outline the most innovative and important arguments and core contents in your paper to attract readers.
 - See annotated document. Thank you.
- Audio core tip
 - Attached as requested. Thank you.
- Please reformat all the reference numbers like this. Please check throughout.
 - See annotated document. Thank you.
- Please write the COMMENTS section at here. See the format in the Format.
 - See annotated document. Thank you.
- Please add PubMed citation numbers and DOI citation to the reference list and list all authors. Please revise throughout. The author should provide the first page of the paper without PMID and DOI.
 - See annotated document and see attached PDFs. Thank you.